# Notes East Area Project Meeting 32 - 22/11/19

Agenda: https://indico.cern.ch/event/851215/

Participants: Sebastien Evrard EN-EA, Eva Montbarbon EN-EA, Adrian Miranda Fontan TE-MPE, Gurkan Dogru EN-EA, Eva Sanchez- Corral BE-ICS, Aboubakr Ebn Rahmoun EN-EA, Jaime Renedo Anglada TE-MSC, Laure Wilhelm EN-SMM, Robert Froeschl HSE-RP, Elpida Iliopoulou HSE-RP, Bastien Rae EN-EA, Louis-Frederic Andre BE-ASR, Evelyne Dho EN-DI, Leszek Borakiewicz IT-CS.

#### **NEWS** S. Evrard

Sebastien presented a report of the CSR3 which took place on last Thursday. The full project has been reviewed for the third time with a particular focus on the following topics:

- LS2 schedule and 2021-operation readiness
- Expenditures and budget forecast
- Activities pending final approval

He then showed the budget indicators at mid of November: 83 % of the budget is committed and 50 % charged. Some groups have already overcommitted their allocated budget and no more expenditures will be authorized.

The CSR panels made several recommendations which should be confirmed in a close-out meeting planned in early December. To be noted that the B.251 new ventilation installation has been put on hold. As a consequence, the Project management decided to keep the existing installation (4 AHUs and 8 heating fans) as it is and to postpone its dismantling.

## Worksite news (B251) G. Dogru

Gurkan reviewed the status of B.251 activities. The air duct installation is now completed. In November/December, the following activities are planned:

- EN/CV: Install Demin water circuit
- EN/EL: install cable trays near the transformers, install cable trays in B.251 as from Dec. 16
- SMB (ALTEAD): deliver False floor structure on Dec 06 and start install it on Dec. 09

## Worksite news (B352+157) A. Ebn Rahmoun

Aboubakr reported the status of B.251 activities. The dismantling of the primary area has very well progressed in the last weeks:

- The primary area has been declassified in supervised area (previously controlled area)
- De-cabling EL and services completed

In B. 157, the core drilling (sample of concrete) is done, SMB also finished to install sky domes. EN-CV removed water & air pipes in the galleries and on North and South walls. Cloud run is still on-going and the Cloud team is preparing for the modification on the Exp. Area to start in December. The survey team completed the check of the IRRAD&CHARM beam height and resumes ground marking of the shielding blocks location. IT and Swisscom team moved the GSM network from South shielding to South wall. EN-EL started the transformer & star point installation along the North Wall.

In B.352, PS – F61, the status is:

- Cleaning & painting floor completed (rest corridor and below blocks)
- Ground Marking done, (need to reschedule Q74 marking because the slabs have been moved to check their correct stability (SMB&HE))
- Magnet supports ready@b.157
  - Installation of first supports ongoing
- EN/CV, installation of water pipe for the magnets, on going
- TE/MSC&EA, preparation of magnets installation
- EN/CV, ACE, HE&EA: a lot of co-activities during November-December
- EN/HE&EA check methods of MCB magnet installation

## Schedule B. Rae

Bastien presented the detailed schedule for the 6 next months emphasizing that a crash program has been established to recover the identified delay for installing the B.251 false-floor at a cost of more co-activities (3months  $\rightarrow$  1 month). Evelyne asked when the dismounting of the Cloud metallic platform will happen: Bastien will add this activity which is currently not included in the planning: only the installation of the new one is shown. The baseline is to have the removal of the former one just before the installation of the new one.

## Safety issues L-F Andre

Louis-Frederic reported on the recent safety event (cut of an old phone cable), reminded the importance of wearing PPEs especially as a lot of co-activities are going to take place in the coming months. He also reported on the CSAP meeting in October where he presented the safety status of the project.

#### **AOB**

No AOB to discuss.

Next meeting on December 13<sup>th</sup> at 15.00.

SEV 25.11.19